Author's response to reviews

Title: Multiple Causes for Delay in Arrival at Hospital in Acute Stroke Patients in Aydin, Turkey

Authors:

Sakine Memis (smemis@adu.edu.tr)
Emel Tugrul (emelturul@yahoo.com)
Emine Didem Evci (devci@yahoo.com)
Filiz Ergin (fergin@adu.edu.tr)

Version: 5 Date: 12 February 2008

Author's response to reviews: see over
Dear Editor,

We wish to publish our article named *Multiple Causes for Delay in Arrival at Hospital in Acute Stroke Patients in Aydin, Turkey* in the BMC Neurology. The article is original, has not been previously published and is not for consideration by another journal. It has been prepared according to the manuscript rules. We believe that this article would be helpful to health care decision makers as they develop a model for stroke health care and community based training.

All of the authors have read the article and approved the content. The authors transfer all copyright ownership of the manuscript entitled *Multiple Causes for Delay in Arrival at Hospital in Acute Stroke Patients in Aydin, Turkey*, in the event the work is published. There have been any commercial associations that could pose a conflict of interest or financial bias during the study.

Please find attached author’s answers to reviewers.

Sincerely,

Asist. Prof.Dr. Sakine MEMIS
Department of Medical Nursing, Adnan Menderes University, School of Health, Aydin-Turkey
Tel: Mobile: 0 552 685 21 34
Work: +090 256 214 80 66/124
Fax: +090 256 212 31 69
e-mail: smemis@adu.edu.tr
Postal Address
Girne Mahallesi Yoruk Ali Efe Bulvari
Ermekan Sitesi, L-Blok No: 1
09100 Aydin TURKEY